\begin{thebibliography}{10}

\bibitem{Agarwal91}
A.~Agarwal, D.~Chaiken, G.~D'Souza, K.~Johnson, D.~Kranz, J.~Kubiatowicz,
  K.~Kurihara, B.~H. Lim, G.~Maa, D.~Nussbaum, M.~Parkin, and D.~Yeung.
\newblock {The MIT Alewife Machine: A Large-Scale Distributed-Memory
  Multiprocessor}.
\newblock In {\em Proceedings of Workshop on Scalable Shared Memory
  Multiprocesors}. Kluwer Academic Publishers, 1991.

\bibitem{StarTNG94}
B.~S. Ang, Arvind, and D.~Chiou.
\newblock {{StarT the Next Generation: Integrating Global Caches and Dataflow
  Architecture}}.
\newblock In {\em Proceedings of the ISCA '92 Dataflow Workshop, Hamilton
  Island, Australia (to be published)}, May 1992.

\bibitem{CCDSM94}
B.~S. Ang, D.~Chiou, and Arvind.
\newblock {Issues in Building a Cache-Coherent Distributed Shared Memory
  Machine using Commercial SMPs}.
\newblock {CSG} Memo 365, {Laboratory for Computer Science, MIT}, Cambridge MA,
  December 1994.

\bibitem{arvind90}
Arvind and R.~S. Nikhil.
\newblock {Executing a Program on the {MIT} Tagged-Token Dataflow
  Architecture}.
\newblock {\em IEEE Transactions on Computers}, 39(3):300--318, March 1990.

\bibitem{BlumofeJoKuLeRaShZh94}
R.~D. Blumofe, C.~F. Joerg, B.~C. Kuszmaul, C.~E. Leiserson, K.~H. Randall,
  A.~Shaw, and Y.~Zhou.
\newblock {C}ilk: An efficient multithreaded runtime system.
\newblock Submitted for publication, December 1994.
\newblock Available via anonymous FTP from {\small\tt theory.lcs.mit.edu} in
  {\small\tt /pub/cilk/cilkpaper.ps.Z}.

\bibitem{Boughton94}
G.~A. Boughton.
\newblock {Arctic Routing Chip}.
\newblock In {\em {Parallel Computer Routing and Communication: Proceedings of
  the First International Workshop, PCRCW '94}}, volume 853 of {\em Lecture
  Notes in Computer Science}, pages 310--317. Springer-Verlag, May 1994.

\bibitem{Dally92}
W.~J. Dally and et~al.
\newblock {Architecture of a Message-Driven Processor}.
\newblock {\em IEEE Micro}, 12(2):23--39, 1992.

\bibitem{Dwarkadas93}
S.~Dwarkadas, P.~Keleher, A.~L. Cox, and W.~Zwaenepoel.
\newblock {Evaluation of Release Consistent Software Distributed Shared Memory
  on Emerging Network Technology}.
\newblock In {\em 4th. ACM Symposium on Principles and Practice of Parallel
  Programming (PPoPP)}. ACM, 1993.

\bibitem{Hagersten92}
E.~Hagersten.
\newblock {\em {Toward Scalable Cache Only Memory Architectures}}.
\newblock PhD thesis, The Royal Institute of Technology, Stockholm, Sweden,
  October 1992.

\bibitem{Heinrich94}
M.~Heinrich, J.~Kuskin, D.~Ofelt, J.~Heinlein, J.~Baxter, J.~P. Singh,
  R.~Simoni, K.~Charachorloo, D.~Nakashira, M.~Horowitz, A.~Gupta,
  M.~Rosenblum, and J.~Hennessy.
\newblock {The Performance Impact of Flexibility in the Stanford {FLASH}
  Multiprocessor}.
\newblock In {\em Proceedings of the Sixth International Conference on
  Architecture Support for Programming Languages and Operating Systems, San
  Jose, CA}, pages 274 -- 285, October 1994.

\bibitem{Joerg90}
C.~F. Joerg.
\newblock Design and implementation of a packet switched routing chip.
\newblock Technical Report MIT/LCS/TR-482, Laboratory of Computer Science,
  Massachusetts Institute of Technology, December 1990.

\bibitem{Keckler&Dally92}
S.~W. Keckler and W.~Dally.
\newblock {Processor Coupling: Integrating Compile Time and Runtime
  Scheduling}.
\newblock In {\em Proceedings of the 19th Annual International Symposium on
  Computer Architecture}, pages 202--213. ACM, 1992.
\newblock ISCA '92-Gold Coast, Australia.

\bibitem{Lenoski90}
D.~Lenoski, J.~Laudon, K.~Charachorloo, A.~Gupta, and J.~Hennessy.
\newblock {The Directory-Based Cache Coherence Protocol for the DASH
  Multiprocesor}.
\newblock In {\em Proceedings of The 17th Annual International Symposium on
  Computer Architecture, Seattle, WA}, pages 148--159, 1990.

\bibitem{Lenoski&others92}
D.~Lenoski, J.~Laudon, T.~Joe, D.~Nakahira, L.~Stevens, A.~Gupta, and
  J.~Hennessy.
\newblock {The DASH Prototype: Implementation and Performance}.
\newblock In {\em Proceedings of The 19th Annual International Symposium on
  Computer Architecture, Gold Coast, Australia}, pages 92--103, May 1992.

\bibitem{Li86}
K.~Li.
\newblock {\em Shared Virtual Memory on Loosely Coupled Multiprocessors}.
\newblock PhD thesis, Yale University, September 86.
\newblock (Also as YALE/DCS/RR-492).

\bibitem{Li89a}
K.~Li, M.~Stumm, D.~Wortman, and S.~Zhou.
\newblock {Shared Virtual Memory Accommodating Hetergeneity}.
\newblock CS-TR 210-89, Princeton University, Department of Computer Science,
  February 1989.

\bibitem{Id}
R.~S. Nikhil.
\newblock {Id Reference Manual, Version 90.1}.
\newblock {CSG} Memo 284-2, {Laboratory for Computer Science, MIT}, Cambridge
  MA, September 1990.

\bibitem{Nikhil94}
R.~S. Nikhil.
\newblock {Cid: A Parallel "Shared-memory" C for Distributed-memory Machines}.
\newblock In {\em {Proceedings of the 7th Annual Workshop on Languages and
  Compilers for Parallel Computing, Ithaca, NY}}, Lecture Notes in Computer
  Science. Springer-Verlag, August 1994.

\bibitem{Papadopoulos91}
G.~M. Papadopoulos.
\newblock {\em Implementation of a General-Purpose Dataflow Multiprocessor}.
\newblock The MIT Press, 1991.
\newblock Research Monograph in Parallel and Distributed Computing.

\bibitem{Papadopoulos93}
G.~M. Papadopoulos, G.~A. Boughton, R.~Greiner, and M.~J. Beckerle.
\newblock {*T: Integrated Building Blocks for Parallel Computing}.
\newblock In {\em Proceedings of Supercomputing '93, Portland, Oregon}, pages
  624--635, November 1993.

\bibitem{Reinhardt93}
S.~K. Reinhardt, M.~D. Hill, J.~R. Larus, A.~R. Lebeck, J.~C. Lewis, and D.~A.
  Wood.
\newblock {The Wisconson Wind Tunnel: Virtual Prototyping of Parallel
  Computers}.
\newblock In {\em ACM SIGMETRICS}, May 1993.

\bibitem{Reinhardt94}
S.~K. Reinhardt, J.~R. Larus, and D.~A. Wood.
\newblock {Tempest and Typhoon: User-Level Shared Memory}.
\newblock In {\em Proceedings of the 21st Annual International Symposium on
  Computer Architecture, Chicago, Il}, pages 325--336, April 1994.

\bibitem{Sakai89}
S.~Sakai, Y.~Yamaguchi, K.~Hiraki, Y.~Kodama, and T.~Yuba.
\newblock {An Architecture of a Dataflow Single Chip Processor}.
\newblock {\em Proceedings of the 16th Annual International Symposium on
  Computer Architecture, Jerusalem, Israel}, pages 46--53, 1989.

\end{thebibliography}

%%=========================================================================
% * Overall goal: overview of \start/, the design philosopy, and 
%                 justification.  
%   - building a machine out of as much stock components, both hardware
%     and software as possible.  
%   - reason for this is:
%     # high rate at which commercial microprocessors are improving
%     # engineering of a system still takes much effort 
%     # development of systems software take a trememdous amount of
%       effort.
%   - Will be good if we can ``layer'' a parallel machine on top of
% stock machines.  Layering of both hardware and OS, maybe even the
% compiler.  
%   - encourage introduction of parallel processing in an evolutionary
% manner.  
%   - Want to do it in a way that provides the best possible MPP
% support: high bandwidth, low latency, user level network access;
% shared memory support.  
%   - The hope is that eventually, the good features will influence the
% design of microprocessors, e.g. the NIU may make it on chip.   
% 
%%=========================================================================

%%------------------------------------------------------------------------
\section{Introduction}
%%------------------------------------------------------------------------
\label{sec:intro}
%%------------------------------------------------------------------------


The past few years have seen the demise of many companies dedicated to
making high performance parallel computers.  Some members of the
computing community have gone as far as saying that parallel
processing, in a classic sense, is dead.  Although we strongly
disagree with this assessment, we do agree that parallel computing is
still at an adolescent stage in its development.  We believe the
problem is two-fold; it is too hard to program parallel computers, and
the hardware, especially for {\em massively} parallel machines, costs
too much and does too little.  We are trying to solve the first
problem by using implicitly parallel functional languages like
Id\cite{Id,arvind90} and pH, and multithreaded languages such as
Cid\cite{Nikhil94} and Cilk\cite{BlumofeJoKuLeRaShZh94}.  This paper,
however, concentrates on \start/, our solution to the second issue.

Mainstream computers have become commodities.  PC's have been selling
in the millions, forcing down prices, speeding up time tables and
making rapid strides in performance.  The economies of scale required
to stay competitive have resulted in consortia to define and design
common systems, enabling hardware to be manufactured in huge
quantities.  Parallel computers, on the other hand, have traditionally
used a lot of custom hardware necessitating the use of custom
software.  By the time the machine is ready to be sold, its processing
node is generally one or two generations out-of-date, making the
current commercial microprocessors a factor of two or more faster.
Since both the development team and customer base are small, problems
cannot be found and worked out quickly, often making custom machines
and their software unreliable.

Coupling unreliability with the high cost of custom development, the lack
of shrink-wrapped software due to the unique nature of parallel machines
and the difficulty in writing custom applications, buying a parallel
computer is difficult to justify.  One would buy such a system only if the
application was critical enough to warrant a dedicated {\em expensive}
machine and the associated custom software development and maintenance
cost.  Massively parallel computers have fallen into the class of
traditional supercomputers, rather than being affordable, widely-available
high performance computers as originally envisioned.

\start/\footnote{The {\sc *T} or {\sc StarT} project has gone through
many changes over its lifetime.  For a reasonably detailed history of
the many generations of {\sc *T}, see \cite{StarTNG94}}, a joint
project between MIT and Motorola, tries to address these problems.
\start/ is based on a commercial symmetric multiprocessor (SMP) {\em
system} using PowerPC 620 processors. The goal of the project is to
deliver very aggressive parallel performance by making small,
manageable changes to the base SMP.  We have added support for low
overhead, high-bandwidth, user-level messaging, and support for
globally coherent shared memory.  Starting from a commercial system
allows us to leverage infrastructure such as the processor, operating
system, memory subsystem, and I/O subsystem.  By borrowing most of the
system technology, we dramatically reduce development time and cost,
allowing us to deliver \start/ at approximately the time the base SMP
is introduced.  \start/, though a research machine, is {\em
commercially} competitive in parallel performance.  \start/ can also
run stock sequential applications efficiently.  Just as SMP's allow
better sharing of resources such as processors, memory and I/O
capability between jobs, \start/ allows better sharing of a collection
of such resources by being able to run sequential, shared-memory and
message-passing parallel applications.

While there are many advantages to using an entire system as the building
block of a parallel machine, there are many technical challenges as well.
Our design not only has to live with the PowerPC 620 microprocessor, which
is optimized for sequential execution, it cannot change the base system in
any significant way.  Observing these tight constraints while providing
competitive performance is the topic of this paper.

{\bf Organization:} The next section makes a slight detour to explore
the next generation of microprocessors.  Understanding certain
processor issues is essential to understanding \start/.  In
Section~\ref{sec:hdw-overview}, we present an overview of the \start/
hardware.  This is followed in Section~\ref{sec:msg-support} with a
discussion of message passing support on \start/.
Section~\ref{sec:sm-support} discusses how shared memory is
implemented on \start/.  Finally, we compare \start/ with some related
work in Section~\ref{sec:related-wk} before concluding with the
current status of the machine.

%%------------------------------------------------------------------------
%%                               END
%%------------------------------------------------------------------------

